Author’s response to reviews

Title: Endometrial carcinoma in a gravid uterus: A case report and literature review

Authors:

Mayu Shiomi (mayu.shiomi@gmail.com)
Shinya Matsuzaki (zacky_bm@gyne.med.osaka-u.ac.jp)
Eiji Kobayashi (ekobayashi@gyne.med.osaka-u.ac.jp)
Takeya Hara (tttake.0303@gmail.com)
Satoshi Nakagawa (s.nakagawa@gyne.med.osaka-u.ac.jp)
Tsuyoshi Takiuchi (takkitakkitakki3@gmail.com)
Kazuya Mimura (mimura@gyne.med.osaka-u.ac.jp)
Yutaka Ueda (y.udea@gyne.med.osaka-u.ac.jp)
Takuji Tomimatsu (tomimatsu@gyne.med.osaka-u.ac.jp)
Tadashi Kimura (tadashi@gyne.med.osaka-u.ac.jp)

Version: 2 Date: 31 Aug 2019

Author’s response to reviews:

August 31, 2019

Tovah Honor Aronin, Ph.D

Editor

BMC Pregnancy and Childbirth

RE: PRCH-D-19-00764_R2

Dear editor:

Thank you for reviewing our manuscript titled “Endometrial carcinoma in a gravid uterus: A case report and literature review.” The editor’s and reviewers’ comments were very helpful, and
we have revised our manuscript accordingly. Our point-by-point responses to the reviewer’s comments are provided below.

We would like to take this opportunity to express our sincere gratitude to the reviewers who identified areas of the manuscript that needed corrections or modifications. We would also like to thank you for allowing us the opportunity to resubmit a revised copy of the manuscript.

We trust that the revised manuscript will now be suitable for publication in BMC Pregnancy and Childbirth

Sincerely,

Dr Shinya Matsuzaki
Department of Obstetrics and Gynaecology
Osaka University Graduate School of Medicine
2-2 Yamadaoka, Suita, Osaka 565-0871, Japan
Phone No.: +81-6-6879-3355
Fax No.: +81-6-6879-3359
Email address: zacky@gyne.med.osaka-u.ac.jp

Editor’s comments

Thank you very much for your helpful comments. We have responded to the comments and explanations regarding revisions made to the manuscript.

1. Please address the final reviewer's comments at the end of this email.

According to the reviewer’s suggestion, we have added the limitation of our study to lines 173 through 176.
2. Please rename the Discussion to 'Discussion and Conclusion'.

As you suggested, we have revised on line 142.

3. In your Competing Interests section, please state that author Shinya Matsuzaki is an Associate Editor for BMC Pregnancy and Childbirth.

Thank you for your helpful comment. We have added the description on line 231.

4. Please confirm who AK is in your Author’s contributions, as these initials do not match any of the authors in the author list.

We apologize for our mistake. We have deleted “AK”.

5. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

As you suggested, we have uploaded the clean files.

Reviewer’s comments

Reviewer #3:

REQUESTED REVISIONS:

The authors may need to address, as a limitation, if they think that the reporting of the literature might be biased by including only published literature and cases that made it to the scientific publication stage and this condition might be under-reported.

We appreciate your insightful comments. According to your suggestion, we have added the limitation of our study to lines 173 through 176.